Cognition and Behavior

# Complex Interactions between Genes and Social Environment Cause Phenotypes Associated with Autism Spectrum Disorders in Mice

### <sup><ao</sup>Monika Sledziowska, Shireene Kalbassi, and <sup><ao</sup>Stéphane J. Baudouin

https://doi.org/10.1523/ENEURO.0124-20.2020

School of Biosciences, Cardiff University, Cardiff CF10 3AX, United Kingdom

### Abstract

The etiology of autism spectrum disorders (ASDs) is a complex combination of genetic and environmental factors. Neuroligin3, a synaptic adhesion protein, and cytoplasmic *FMR1* interacting protein 1 (CYFIP1), a regulator of protein translation and actin polymerization, are two proteins associated with ASDs that interact in neurons *in vivo*. Here, we investigated the role of the Neuroligin3/CYFIP1 pathway in behavioral functioning and synapse formation in mice and found that it contributes to motor learning and synapse formation in males. Similar investigation in female mice revealed an absence of such phenotypes, suggesting that females are protected against mutations affecting this pathway. Previously, we showed that the social environment influences the behavior of male mice. We extended this finding and found that the transcriptome of wild-type mice housed with their mutant littermates, lacking Neuroligin3, differed from the transcriptome of wild-type mice housed together. Altogether, these results identify the role of the Neuroligin3/CYFIP1 pathway in male mouse behavior and highlight its sensitivity to social environment.

Key words: autism spectrum disorders; motor learning; mouse behavior; social environment

### Significance Statement

The causes of autism spectrum disorders (ASDs) remain elusive. In this study, we investigate the combined effect of mutations in two genes associated with ASDs, *NIgn3* and *Cyfip1*, and the effect of the social environment on phenotypes relevant for ASDs. We show that when both mutations are present the behavior can be restored, emphasizing the importance of considering gene interactions. We also show sex differences in behavior, suggesting that female subjects should be included in the studies of ASDs. We show that wild-type animals can exhibit phenotypes associated with ASDs as a result of being housed with their mutant littermates, highlighting the necessity to re-evaluate the use of wild-type animals as controls to define phenotypic traits of mouse models.

### Introduction

Autism spectrum disorders (ASDs) are characterized by behavioral manifestations, primarily social communication deficits and stereotyped behavior, frequently accompanied by a wide range of secondary comorbid clinical symptoms (American Psychiatric Association, 2012). The etiology of ASDs is far from being fully understood but is likely to be a complex interaction between specific

Received March 31, 2020; accepted June 25, 2020; First published July 15, 2020.

The authors declare no competing financial interests.

genetic mutations and environmental factors triggering the emergence of a constellation of behavioral symptoms. Understanding the relationship between the different causes of the disorder and the behavioral manifestation is key for the development of targeted treatments.

Among the genes associated with ASDs is the X-linked gene *NLGN3* (Jamain et al., 2003; Talebizadeh et al., 2006; Glessner et al., 2009; Yasuda et al., 2011; Sanders

Author contributions: M.S. and S.J.B. designed research; M.S. and S.K. performed research; M.S. and S.J.B. analyzed data; M.S. and S.J.B. wrote the paper.

et al., 2015; Gumus, 2019), coding for the synaptic adhesion molecule Neuroligin3 (Chih et al., 2004; Varoqueaux et al., 2006) and cytoplasmic FMR1 interacting protein 1 (CYFIP1; Wang et al., 2009; van der Zwaag et al., 2009; Toma et al., 2014; Picinelli et al., 2016), and coding for a protein involved in actin polymerization regulation (Chen et al., 2010) and the regulation of protein translation (Napoli et al., 2008). In male mice, the lack of Nlgn3 is associated with impaired social behavior (Radyushkin et al., 2009; Fischer and Hammerschmidt, 2011; Kalbassi et al., 2017; Bariselli et al., 2018), hyperactivity (Radyushkin et al., 2009), and subtle changes in the rate of motor learning (Baudouin, 2014; Rothwell et al., 2014). In mice, Cyfip1 haploinsufficiency is also associated with reduced interest in social odors, hypoactivity, and motor learning impairment (Bozdagi et al., 2012; Bachmann et al., 2019). Interestingly, the cytoplasmic tail of Neuroligin3 contains a WAVE regulatory complex interactor receptor sequence, which can bind CYFIP1 (Chen et al., 2014; Bachmann et al., 2019), an interaction that was recently confirmed in neurons in vivo (Bachmann et al., 2019). The interaction between Neuroligin3 and CYFIP1 suggests that they are part of the same molecular pathway and that disruption of this pathway can cause some of the phenotypes associated with ASDs.

At the subcellular level, in neurons, both proteins have been associated with the formation and elimination of synapses. Triple knockout of Nlgn3 as well as two other members of this gene family, Nlgn1 and Nlgn2, results in a reduction of synaptic contacts in vivo and lower dendritic spine number in vitro (Chih et al., 2004; Varoqueaux et al., 2006), supporting the idea that *NIgn3* might play a role in synaptic development and maintenance. Cyfip1 haploinsufficiency is associated with a decrease in spine density in the motor cortex and olfactory bulb in male mice (Abekhoukh et al., 2017; Bachmann et al., 2019) as well as an increased number of filamentous spines (De Rubeis et al., 2013; Pathania et al., 2014; Abekhoukh et al., 2017; Davenport et al., 2019). This is potentially explained by the increased rate of formation and elimination of dendritic spines seen in these animals (Bachmann et al., 2019). Importantly, the alteration of dendritic spine density is one of the few cellular phenotypes of ASDs observed both in mouse models and in human patients (Martínez-Cerdeño, 2017). The overall aim of our study was to investigate how the interaction between *Nlgn3* and *Cyfip1* can influence ASD-relevant phenotypes in mice, from behavioral alterations to changes at the cellular level.

This research was supported by the Sêr Cymru program of the Welsh Government; and Cardiff University Grant 501100000866 and Wellcome Trust Grant 100010269 to M.S. and S.J.B.

Acknowledgments: We thank Angela Marchbank at the Cardiff University Genome Hub for performing the RNA sequencing and Daniel Pass for help with the analysis.

Correspondence should be addressed to Monika Sledziowska at sledziowskam@cardiff.ac.uk.

https://doi.org/10.1523/ENEURO.0124-20.2020

Copyright © 2020 Sledziowska et al.

The complex genotype-phenotype relationships are further shaped by factors such as sex of the subjects and their social environment. For example, only females with the complete deletion of Nlgn3 showed deficits in behavior (Kalbassi et al., 2017). The social environment might also be an environmental factor triggering ASD-related phenotypes. Male mice lacking Nlgn3 and wild-type (WT) mice influence each other's behavior, an effect attributed to the role of *Nlgn3* in controlling social dominance. Housing mice carrying the 16p11.2 microdeletion, another model of ASD, with littermates of a different genotype was also shown to influence the vocalization during courtship (Yang et al., 2015). In addition to investigating the role of interaction between NIgn3 and Cyfip1 in controlling behavioral phenotypes, the secondary aims of this study were to determine the role of sex in the phenotypic outcome of mutations in NIgn3 and Cyfip1 and the effect of social environment on mRNA expression in these mice.

### **Materials and Methods**

#### Animals

All procedures were conducted in accordance with the Animals (Scientific Procedure) Act 1986 (amended in 2012). Mice were kept on a 12 h light/dark cycle with free access to food and water, in groups of two to five in a cage. All behaviors were assessed during the light phase of the light/dark cycle. All mice were tested as adults, P60–P70 at the start of testing. Mice were handled for at least 2 days before any procedure. The mice were habituated to the room where the behavioral assessment was taking place for 30 min before commencing any procedure.

 $NIgn3^{y/-}$ ,  $NIgn3^{+/-}$  (Tanaka et al., 2010) and  $Cyfip1^{+/-}$ (EUCOMM) mice were crossed with mice containing Thy-EGFP transgene (stock # 007788, The Jackson Laboratory; Feng et al., 2000) to obtain the following male mice: Cyfip1<sup>+/-</sup>Thy1-EGFP, Nlgn3<sup>y/-</sup> Nlgn3<sup>y/-</sup>Thy1-EGFP, Cyfip1+/-Thy1-EGFP, and Thy1-EGFP; and the following female mice: Nlgn3<sup>+/-</sup>Thy1-EGFP, Cyfip1<sup>+/-</sup>Thy1-EGFP, Nlgn3<sup>+/-</sup>Cyfip1<sup>+/-</sup>Thy1-EGFP, and Thy1-EGFP. Thus, all of the mice used in the behavioral experiments and used to investigate the dendritic spine density were littermates. A proportion of the mice used in the behavioral experiments had the Thy-EGFP transgene. The lack of an effect of the transgene on the behavioral outcomes was confirmed by repeating every statistical analysis with the presence of the transgene as a variable. There were no significant differences in any of the behavioral outcomes between the mice with and without the transgene.

In the first RNA experiment, an additional cohort of wild-type animals was bred and included in the analysis. In the second RNA experiment, a cohort of wild-type animals was bred where the parents came from the *NIgn3* colony, ensuring that the mutant and wild-type mice shared parents.

#### Activity in open field

The spontaneous activity of mice was recorded. Mice were tested on 2 consecutive days. During the test, the mice were individually placed in an open field arena (40  $\times$ 

This is an open-access article distributed under the terms of the Creative Commons Attribution 4.0 International license, which permits unrestricted use, distribution and reproduction in any medium provided that the original work is properly attributed.

20 cm) and were allowed to explore. The test was conducted in the dark; however, the bottom of the arena was illuminated by an infrared lamp to allow tracking of the mice. The movement of the mice was recorded using a video camera above the arena. The traces were recorded and quantified in EthoVision XT (Noldus).

### Rotarod

Motor learning of the mice was assessed using a rotarod (Jones and Roberts, 1968). Latency to fall off the rod was assessed for 3 days in a row, with 10 subsequent 5 min trials each day. During a trial, mice were placed on the rod. The rotarod was then switched on and accelerated from 4 to 40 rpm over the course of 5 min. The mice were allowed to walk on the rod until they fell off, gripped to the rod and the rod made a full revolution, or 5 min have passed. Falling off or gripping the rod was interpreted as an inability to cope with the task and signaled the end of the trial. Latency to fall was measured using a stopwatch (Casio). After each trial, the mice were allowed to rest for 5 min at the bottom of the apparatus.

### Social odor interest

Social odors originated from a cage of three to four WT mice that were maintained with the same home cage bedding for a week to allow for the concentration of odorants present in the urine. For some of the trials, the cage also contained a maximum of one  $NIgn3^{y/-}Cyfip1^{+/-}$  or  $Nlgn3^{+/-}Cyfip1^{+/-}$  mouse. Before the trial, a cotton bud was wiped across the bottom of the home cage in a zig-zag fashion to obtain the social odor cue. Mice were placed in the experimental arena and were allowed to habituate for 2 min. Mice were exposed to a clean cotton bud for 2 min, which was then swapped for a new clean cotton bud, and mice were allowed to interact with it for another 2 min. Next, the mice were exposed to the cotton bud with the odor cue for 2 min, which was finally swapped for a new cotton bud with an olfactory cue for another 2 min. Male mice were exposed to olfactory cues originating from a cage of male mice, while female mice were exposed to olfactory cues originating from a cage of female mice. The trials were recorded with a video camera placed above the experimental arena. Time spent sniffing the cotton bud was scored manually, blinded to the genotype.

### **Courtship vocalization**

Female mice in estrus were identified using vaginal lavage, followed by cytological staining (Giemsa solution, Polysciences) and visual assessment. Male mice were habituated to the experimental arena ( $40 \times 20$  cm) for 3 min. Next, an unfamiliar female mouse in estrus was added to the arena, and the mice were allowed to interact freely for 3 min. Ultrasonic vocalizations (USVs) between 40 and 250 Hz produced by the male mice were recoded using a preamplifier (UltraSoundGate 416 H, Avisoft Bioacoustics) connected to a microphone (UltraSoundGate CM16, Avisoft Bioacoustics). The total number of USVs and their duration was analyzed using SASLabPro (Avisoft Bioacoustics).

### Histology

Dendritic spine quantification was conducted in the motor and visual cortices of Nlgn3<sup>y/-</sup>Cyfip1<sup>+/-</sup>, Nlgn3<sup>y/-</sup>, Cyfip1<sup>+/-</sup> males and their WT littermates, as well as of  $Nlgn3^{+/-}Cyfip1^{+/-}$ ,  $Nlgn3^{+/-}$ , and  $Cyfip1^{+/-}$  females and their WT littermates, all of which also expressed EGFP under the Thy-1 promoter. Mice were anesthetized with Euthatal and perfused with 4% paraformaldehyde in 0.1 м PBS. The entire brain was dissected out and postfixed overnight in the 4% paraformaldehyde in 0.1 M PBS, kept in 30% sucrose solution until saturated and stored at -80°C. The brains were cut coronally into 50  $\mu$ m sections on a cryostat (Leica Biosystems) and immediately mounted on glass slides. The regions of interest were identified using a mouse brain atlas (Paxinos and Franklin, 2004), and Z-stack images spaced 0.5 mm apart were acquired on a Zeiss LSM700 upright confocal microscope (Carl Zeiss), using a 40 waterimmersion lens. The images were reconstructed into two dimensions using Z-stack maximum intensity projection in ImageJ (NIH). The images were analyzed by an experimenter blinded to the genotype of the animals. Spines were identified manually and counted on a 20- to 250-µm-long stretch of a dendrite, with a minimum of 24 dendrites from four mice, per condition. Spine density was calculated as the number of spines per 10  $\mu$ m of a dendrite.

### **RNA** sequencing

All procedures were conducted in RNAase-free conditions. Mice were culled by cervical dislocation, their brains were extracted, and relevant brain regions were dissected following a mouse brain atlas (Paxinos and Franklin, 2004). The samples were immediately snap frozen in liquid nitrogen and stored at -80°C. Invitrogen TRIzol Reagent (Thermo Fisher Scientific) was added to the samples, at 1 ml per 50-100 mg of tissue. Tissue was homogenized and incubated at room temperature for 5 min. Samples were transferred to Invitrogen Phrasemaker Tubes (Thermo Fisher Scientific), and 0.2 ml of chloroform was added. The samples were shaken for 15 s, incubated at room temperature for 15 min, and centrifuged for 5 min at 14,000  $\times$  g, at 4°C. The RNA-containing upper phase was mixed with 0.5 ml of isopropanol, and the samples were incubated for 1 h at -80°C. Following the incubation, the samples were centrifuged for 10 min at 10,000  $\times q$ . The supernatant was removed, and the pellet was washed using 75% EtOH. The pellet was dissolved in RNA-free water and treated with DNase (QIAGEN) as per manufacturer instructions.

For reverse transcription, 1250 ng of RNA was incubated with 1  $\mu$ l of random primers (Promega) and 1  $\mu$ l of deoxyribonucleotide triphosphates (dNTPs; 10 mM, Promega) for 5 min, at 65°C and for 1 min on ice. Then 4  $\mu$ l of buffer (Thermo Fisher Scientific), 1  $\mu$ l dithiothreitol (0.1 M, Thermo Fisher Scientific), 1  $\mu$ l Rnasin plus (Promega), and 1  $\mu$ l of superscript III reverse transcriptase (Thermo Fisher Scientific) were added, and the mix was incubated for 5 min at room temperature and then for 2 h at 50°C. The samples were then incubated for 10 min at 70°C.

Quality control of the RNA samples was confirmed by Tape Station and Qubit. The library was prepared



**Figure 1.** The effect of *Nlgn3* deletion and *Cyfip1* haploinsufficiency on exploratory behavior. *A*, Distance traveled in the open field by the *Nlgn3<sup>y/-</sup>Cyfip1<sup>+/-</sup>* male mice and their littermates. *B*, Habituation to the novel environment in the *Nlgn3<sup>y/-</sup>Cyfip1<sup>+/-</sup>* male mice and their littermates in the open field. *C*, Distance traveled in the open field in by the *Nlgn3<sup>+/-</sup>Cyfip1<sup>+/-</sup>* female mice and their littermates. *D*, Habituation to the novel environment in the *Nlgn3<sup>+/-</sup>Cyfip1<sup>+/-</sup>* female mice and their littermates. *D*, Habituation to the novel environment in the *Nlgn3<sup>+/-</sup>Cyfip1<sup>+/-</sup>* female mice and their littermates in the open field. \**p* < 0.05, \*\**p* < 0.01, \*\*\**p* < 0.001, n.s., not significant. Comparison of the time spent in the center of the open field arena and sex is available in Extended Data Figure 1-1. Details of statistical analysis are available in Extended Data Figure 1-2.

according to manufacturer instruction (Illumina TruSeq). Total RNA was purified to remove ribosomal and nonmRNA with magnetic beads. mRNA was then transferred for first-strand cDNA synthesis with superscript II. Next, the second strand of the cDNA was synthesized, and the template was eradicated. Adapters were ligated to the cDNA. The cDNA was then amplified to enrich the libraries and tested on a DNA chip for quality control, and the library size was normalized. The RNA sequencing was performed according to manufacturer instructions using the Illumina NextSeq500 System in  $1 \times 75$  bp cartridge. A strand of cDNA was bound to a docking site, and fluorescent dNTPs were added one at a time.

The sequences were trimmed with Trimmomatic (Bolger et al., 2014) and assessed for quality with FastQC. STAR was used to map the reads onto the Mouse Genome Assembly GRCm38. Transcripts were assigned using Feature Counts (Dobin et al., 2013; Liao et al., 2013). Downstream analysis was performed in R version 2.6.2 (R Core Team, 2019). The DESeq2 package was used for differential gene expression (Love et al., 2014). The relative expression of genes was assessed in pairwise fashion to include all housing and genotype conditions. The values were normalized using the implementation of variance-stabilizing transformation from the DESeq2 package. Principal component analysis of the top 100 genes with the greatest fold expression differences was conducted using the R function procomp. Weighted gene

correlation network analysis (Langfelder and Horvath, 2008) was performed on normalized data, with a power of 5 and a minimum module size of 200.

#### Statistical analysis

The data analysis was conducted using R software, version 3.6.2. (R Core Team, 2019) or when an appropriate package was not available using Graph Pad Prism version 8.3.1. The normality of raw data or residuals was checked using the Shapiro–Wilk test as well as visual inspection of Q-Q plots and histograms. The homogeneity of variance was checked using Levene's test or visual inspection of a plot of residuals versus plotted values, depending on the data type. If the data were deemed to violate the assumption of normality or homogeneity of variance, an appropriate nonparametric test was used. The nonparametric mixed ANOVA was conducted according to Noguchi et al. (2012). The details of normality assessment, the tests used, and the number of samples per group can be found in the statistics table in Extended Data Figure 1-2.

#### Results

# *Nlgn* $3^{y/-}$ *Cyfip* $1^{+/-}$ mice phenocopy the enhanced exploratory behavior of *Nlgn* $3^{y/-}$ males

We generated  $Nlgn3^{y/-}Cyfip1^{+/-}$  double-mutant mice to investigate the effect of the Nlgn3/Cyfip1 pathway on



**Figure 2.** The effect of *Nlgn3* deletion and *Cyfip1* haploinsufficiency on motor learning. *A*, Latency to fall off the rotarod in the *Nlgn3<sup>y/-</sup>Cyfip1<sup>+/-</sup>* male mice and their littermates. *B*, Latency to fall off the rotarod in the *Nlgn3<sup>+/-</sup>Cyfip1<sup>+/-</sup>* female mice and their littermates. \*p < 0.05, \*\*p < 0.01, \*\*\*p < 0.001, \*\*\*\*p < 0.001. Data for the comparison of individual trials and sex are available in Extended Data Figure 2-1.

phenotypes associated with autism: changes in behavior and dendritic spine density. In the first instance, we investigated the behavior of the mutant mice.

As previously reported Nlgn3<sup>y/-</sup> males (n = 24; day 2: mean = 6307.63 cm, SE = 256.25 cm) were hyperactive in the open field compared with their WT littermates (n = 18; day 2: mean = 4962.81 cm, SE = 258.24 cm; one-way ANOVA, main effect of genotype:  $F_{(3,62)}$  = 6.55,  $P = \langle 0.001$ , Tukey's HSD:  $t_{(1,62)}$  = 3.47, p < 0.01; Fig. 1A). Similarly, Nlgn3<sup>V/-</sup>Cyfip1<sup>+/-</sup> mice (n = 12; day 2: mean = 6597.82 cm, SE = 420.93 cm)traveled farther in the open field than their WT littermates (Tukey's HSD:  $t_{(1.62)}$  = 3.53, p < 0.01). The Nlgn3<sup>y/-</sup>Cyfip1<sup>+/-</sup> mice were also hyperactive in relation to their Cyfip1<sup>+/-</sup> littermates (n = 12; day 2: mean = 5227.60 cm, SE = 345.70 cm; $t_{(1.62)}$  = 2.7, p < 0.05). However, there was no evidence of a more pronounced phenotype in the  $Nlgn3^{\nu/-}Cyfip1^{+/-}$  mice as they did not significantly differ from the Nlgn3<sup>y/-</sup> singlemutant mice. The same pattern of results emerged when the effect of genotype on the distance traveled was considered across the two days of testing (mixed-model ANOVA, main effect of genotype:  $F_{(3,62)} = 6.45$ , p < 0.001, main effect of day:  $F_{(1,62)}$  = 35.54, p < 0.001; Tukey's HSD: WT vs  $Nlgn3^{y/-1}$ p < 0.01). The hyperactivity in these mice may arise from a more pronounced response to a novel environment. To investigate this possibility, we considered the habituation pattern between the first and second day of testing. A significant decrease in the distance traveled between the 2 days was only observed in the WT males (Tukey's HSD:  $t_{(2,62)} = -5.30$ , p < 0.001; Fig. 1B), suggesting a habituation deficit in the Nlgn3<sup>y/-</sup>, Cyfip1<sup>+/-</sup>, as well as

 $NIgn3^{y/-}Cyfip1^{+/-}$  mice. The phenotype was unlikely to occur because of differences in anxiety levels as there were no differences between genotypes in the time spent in the center of the open field (Extended Data Fig. 1-1*A*).

Sex is known to impact on the rate of diagnosis of ASDs (Werling and Geschwind, 2013); thus, we aimed to investigate potential sex differences in exploratory behavior. We tested Nlgn3<sup>+/-</sup>Cyfip1<sup>+/-</sup> (n = 17) double-mutant female mice alongside their Nlgn3<sup>+/-</sup> (n = 20), Cyfip1<sup>+/-</sup> (n = 10), and WT (n = 16) littermates. The Nlgn3<sup>+//-</sup> mice as well as the NIgn3<sup>+/-</sup>Cyfip1<sup>+/-</sup> mice were no different to their WT littermates in their exploratory behavior in the open field. However, the  $Nlgn3^{+/-}$  females (day 2: mean = 6129.85 cm, SE=301.30 cm) and Nlgn3<sup>+/-</sup>Cyfip1<sup>+/-</sup> females (day 2: mean = 5059.01 cm, SE = 274.50 cm) were hyperactive in relation to their Cyfip  $1^{+/-}$  littermates (day 2: mean = 4550.05 cm, SE = 309.99 cm; one-way ANOVA, main effect of genotype:  $F_{(3,59)} = 4.64$ , p < 0.001; Tukey's HSD:  $t_{(1,59)} = 3.34$ , p < 0.01, and  $t_{(1,59)} = 2.66$ , p < 0.05, respectively; Fig. 1C). This finding was corroborated when the exploratory behavior across the two days of testing was considered (mixed-model ANOVA, main effect of genotype:  $F_{(3,59)}$  = 4.51, p < 0.001; Tukey's HSD *Nlgn*3<sup>+/-</sup> vs *Cyfip*1<sup>+/-</sup>:  $t_{(1.59)}$  = 3.423, p < 0.01). All of the female mice showed a significant decrease in the distance traveled between the first and second day of testing, characteristic of habituation (WT:  $t_{(1,59)} = 5.61$ , p < 0.001; *Nlgn3*<sup>+/-</sup>:  $t_{(1,59)} = 4.42$ , p < 0.01; *Cyfip1*<sup>+/-</sup>:  $t_{(1,59)} = 3.83$ , p < 0.01;  $Nlan3^{+/-}Cyfip1^{+/-}: t_{(1.59)} = 5.30, p < 0.001; Fig. 1D).$  The subtly altered behavior in the female mice cannot be attributed to differences in anxiety levels, as the time spent in the center of the open field was comparable between the different groups (Extended Data Fig. 1-1B). Finally, to

investigate whether there is a sex difference in the exploratory behavior of these mice, we directly compared the WT and *Cyfip1*<sup>+/-</sup> males and females; however, we found no significant differences in the distance traveled or the time spent in the center of the open field between males and females (Extended Data Fig. 1-1*C*,*D*). There were no significant differences between *Nlgn3*<sup>y/-</sup> males and *Nlgn3*<sup>+/-</sup> females neither in their exploratory behavior in the open field (Extended Data Fig. 1-1*E*,*F*).

While  $Nlgn3^{y/-}Cyfip1^{+/-}$  males phenocopied  $Nlgn3^{y/-}$  males in their hyperactivity, the  $Nlgn3^{+/-}Cyfip1^{+/-}$  females were no different from their WT littermates. There was no evidence, however, that reducing the levels of *Cyfip1* affected the level of activity in the males, suggesting that only *Nlgn3* influences this behavior.

# Reduction of *Cyfip1* level restores motor learning in $Nlgn3^{y/-}Cyfip1^{+/-}$ male mice

The ability to learn new motor routines was evaluated by training mice to stay on an accelerating rotating rod, with multiple trials within a day (Extended Data Fig. 2-1) and over 3 consecutive days (Fig. 2). There was no overall effect of genotype on the latency to fall off the rotarod, averaged across days and trials. However, the performance of mice across the days of testing depended on their genotype, indicating that there might be differences in their learning curves (nonparametric mixed-model ANOVA, genotype  $\times$  day interaction:  $F_{(3.1577)} = 4.61$ , p <0.01; Extended Data Fig. 2-1A). In line with this observation, an increased ability to stay on the rod across multiple days was observed in WT male mice (n = 17; day 1 vs day 2 simple effect:  $t_{(1,16)} = 2.12$ , p < 0.05, day 1 vs day 3 simple effect:  $t_{(1,16)} = 6.28$ , p < 0.001; day 2 vs day 3 simple effect:  $t_{(1,16)} = 2.86$ , p < 0.05), in the *Nlgn3<sup>y/-</sup>* male mice (n = 16; day 1 vs day 2 simple effect:  $t_{(1,15)} = 6.16$ , p < 0.001; day 1 vs day 3 simple effect:  $t_{(1,15)} = 9.14$ , p < 0.001; day 2 vs day 3 simple effect:  $t_{(1,15)} = 3.02$ , p < 0.01), and in the Nlgn3<sup>y/-</sup>Cyfip1<sup>+/-</sup> male mice (n = 12; day 1 vs day 2 simple effect:  $t_{(1,12)} = 5.05$ , p < 0.01; day 1 vs day 3 simple effect:  $t_{(1,12)} = 4.69$ , p < 0.01; Fig. 2A). As previously reported, the *Cyfip1<sup>+/-</sup>* mice (n = 12) did not improve in their ability to stay on the rod across the days of training, suggesting that they are unable to learn new motor routines within this protocol (Bachmann et al., 2019). These results indicate that the deficit in motor learning seen in Cyfip1<sup>+/-</sup> mice is restored by deleting NIgn3 in NIgn3<sup>y/-</sup>Cyfip1<sup>+/-</sup> double-mutant mice.

Next, we investigated whether a sex difference was present in the ability of these mice to acquire knowledge of new motor routines. To verify this possibility, we included  $Nlgn3^{+/-}Cyfip1^{+/-}$  female mice and their littermates in the rotarod training. The genotype did not affect the training across the 3 days, suggesting that the learning curves for all of the females were comparable (Extended Data Fig. 2-1*B*). In line with this observation, an increase in time spent on the rod was observed for WT mice (n = 13, day 1 vs day 2:  $t_{(1,13)} = 5.09$ , p < 0.01; day 1 vs day 3:  $t_{(1,13)} = 5.85$ , p < 0.01),  $Nlgn3^{+/-}$  mice (n = 13; day 1 vs day 2:  $t_{(1,13)} = 6.15$ , p < 0.01; day 1 vs day 3:  $t_{(1,13)} = 8.06$ , p < 0.001; day 2 vs day 3:  $t_{(1,13)} = 4.23$ ,

p < 0.05),  $Cyfip1^{+/-}$  mice  $(n = 14; \text{ day } 1 \text{ vs day } 3: t_{(1,13)} = 5.95, p < 0.01)$ , and  $Nlgn3^{+/-}Cyfip1^{+/-}$  mice  $(n = 14; \text{ day } 1 \text{ vs day } 2: t_{(1,14)} = 7.34, p < 0.001; \text{ day } 1 \text{ vs day } 3: t_{(1,14)} = 7.41, p < 0.001; \text{ Fig. } 2B$ ). To determine whether there was a sex difference in the motor learning, we have compared the WT and  $Cyfip1^{+/-}$  males and females directly. On average, females performed better than the males (nonparametric mixed-model ANOVA, main effect of sex:  $F_{(1,52)} = 4.88, p < 0.05$ ; Extended Data Fig. 2-1*C*). Therefore, unlike males, all female mice showed evidence of learning across days. We also compared the *Nlgn3* mutant mice (*Nlgn3^{+/-}* females and *Nlgn3^{y/-}* males) alongside their WT littermates; however, there were no differences in motor learning between the sexes (Extended Data Fig. 2-1*D*).

In male mice, *Nlgn3* and *Cyfip1* act in opposition to control the motor learning on the rotarod. While reducing the level of *Cyfip1* in the males results in impairment in the ability to acquire new motor routines, additional deletion of *Nlgn3* leads to restoration of the behavior. Additionally, sex plays a role in the control of motor learning, where females heterozygous for *Cyfip1* show no deficit.

# Lack of *NIgn3* and *Cyfip1* haploinsufficiency have little effect on social behavior

The social behavior of double-mutant mice and their littermates was evaluated by investigating their interest in social odors and, for the males only, their courtship behavior and direct social interaction with a female in estrus. The ability to discriminate between the control and social odor was observed in WT mice (n = 14; C2; mean = 14.61 s, SE = 1.79 s;S1: mean = 28.46 s; SE = 4.23; simple effect:  $t_{(1,14)} = 4.60$ , p < 0.05), Nlgn3<sup>y/-</sup> mice (n = 16; C2: mean = 14.61 s, SE = 1.79 s; S1: mean = 28.46 s, SE = 4.23 s; simple effect:  $t_{(1,16)} = 5.91$ , p < 0.01), Cyfip1<sup>+/-</sup> mice (n = 12; C2: mean = 12.43 s, SE = 1.72 s; S1: mean = 23.57 s, SE = 1.65 s; simple effect:  $t_{(1,12)} = 8.74$ , p < 0.001), and *Nlgn3<sup>v/-</sup>Cyfip1<sup>+/-</sup>* mice (n = 12; C2: mean = 10.88 s, SE = 2.29 s; S1: mean = 23.62 s,SE=3.54 s; simple effect:  $t_{(1,12)}$ =4.40, p < 0.05; Fig. 3A). However, habituation to the social odor was observed only in in the Cyfip1<sup>+/-</sup> mice (S1: mean = 23.57 s, SE = 1.65 s; S2: mean = 18.40 s, SE = 1.84 s; simple effect:  $t_{(1,12)} = 4.70$ , p < 0.05), and the Nlgn3<sup>y/-</sup>Cyfip1<sup>+/-</sup> mice (S1: mean = 23.62 s, SE = 3.54 s; S2: mean = 16.53 s, SE = 1.858 s; simple effect:  $t_{(1,12)} = 4.71$ ,  $\rho < 0.05$ ). The level of interest in the social odor was also the same between the different genotypes (Fig. 3B).

The genotype of the female mice had no impact on their ability to discriminate between control and social odors. An increase in the time spent with the cue was observed between the control cue and the first presentation of the social odor in WT mice (*n*=13; C2: mean=6.37 s, SE=1.15 s, S1: mean=14.57 s, SE=0.70 s; simple effect:  $t_{(1,13)}=10.80$ , p < 0.001),  $Nlgn3^{+/-}$  mice (*n*=13; C2: mean=6.71 s, SE=1.13 s; S1: mean=12.50 s, SE=0.89 s; simple effect:  $t_{(1,13)}=5.50$ , p < 0.01),  $Cyfip1^{+/-}$  mice (*n*=14; C2: mean=5.96 s, SE=0.69 s; S1: mean=10.57 s, SE=1.33 s; simple effect:  $t_{(1,14)}=4.82$ , p < 0.05), and  $Nlgn3^{+/-}Cyfip1^{+/-}$  mice (*n*=14; C2: mean=7.18 s, SE=1.23 s; S1: mean=14.01 s, SE=1.12 s;



**Figure 3.** *A*, Time spent sniffing the olfactory cue by the  $Nlgn3^{y/-}Cyfip1^{+/-}$  male mice and their littermates. *B*, Time spent sniffing during the first presentation of social olfactory cue depending on the genotype by male mice. *C*, Time spent sniffing the olfactory cue by  $Nlgn3^{+/-}Cyfip1^{+/-}$  female mice and their littermates. *D*, Time spent sniffing during the first presentation of social olfactory cue depending on the genotype by female mice. *E*, Number of ultrasonic vocalisations emitted in response to a female in oestrus by the male mice. *F*, Duration of the ultrasonic vocalisations emitted by male mice. *G*, Time spent in the interaction with a female by male mice. \*p < 0.05, \*\*p < 0.01, \*\*\*p < 0.001.

simple effect:  $t_{(1,14)} = 5.44$ , p < 0.01; Fig. 3*C*). In the female mice, habituation was only observed in *Nlgn3<sup>+/-</sup>* mice (S1: mean = 12.50 s, SE = 0.89 s; S2: mean = 8.34 s, SE = 1.08 s; simple effect:  $t_{(1,13)} = 4.80$ , p < 0.05) and *Nlgn3<sup>+/-</sup>Cyfip1<sup>+/-</sup>* mice (S1: mean = 14.01 s, SE = 1.12 s; S2: mean = 10.10 s, SE = 1.40 s; simple effect:  $t_{(1,14)} = 6.39$ , p < 0.01). The interest in the social odor was reduced in the *Cyfip1<sup>+/-</sup>* females compared with the WT females (one-way ANOVA: main effect of genotype:  $F_{(3,50)} = 2.93$ , p < 0.05; Tukey's HSD:  $t_{(1,50)} = -2.68$ , p < 0.05; Fig. 3*D*).

The courtship behavior of the males exposed to a female in estrus was equivalent in all the mutants (*Nlgn3<sup>v/-</sup>*, n = 16; *Cyfip1<sup>+/-</sup>*, n = 12; *Nlgn3<sup>+/-</sup>Cyfip1<sup>+/-</sup>*, n = 12) and their WT littermates (n = 17). Specifically, the number and the duration of ultrasonic vocalizations emitted in response to the female as well as the time spent in direct interaction with the female was not different between the WT littermates and the other males (Fig. 3*E*-*G*). Deficits in courtship behavior in *Nlgn3<sup>v/-</sup>* males were reported multiple times (Radyushkin et al., 2009; Fischer and Hammerschmidt, 2011; Kalbassi et al., 2017). In

contrast to these findings, there is no evidence for a deficit in courtship vocalization in the *Nlgn3<sup>V/-</sup>* males here. However, it is important to note that while the number and duration of vocalization in the *Nlgn3<sup>V/-</sup>* males are comparable here to the previous literature, the level of vocalization in the WT littermates is lower than expected. The reduced level of vocalization during courtship in the WT males might be a result of being housed with mutant animals, similar to a previously reported effect of the social environment (Kalbassi et al., 2017).

In opposition to previous literature, neither *Nlgn3* nor *Cyfip1* had a substantial impact on social behavior in male mice. This could be because of the effect of the social environment modulating the behavior of the WT controls as well as potentially the mutant animals.

# *NIgn3 and Cyfip1* collectively impact on dendritic spine density in the motor cortex

Altered dendritic spine regulation is another phenotype associated with ASDs (Phillips and Pozzo-Miller, 2015).



**Figure 4.** The effect of *NIgn3* deletion and *Cyfip1* haploinsufficiency on spine density in the cortex. *A*, The dendritic spine density in the M1 area of motor cortex and the V1 area of visual cortex in *NIgn3<sup>y/-</sup>Cyfip1<sup>+/-</sup>* males and their littermates. *B*, The dendritic spine density in the M1 area of motor cortex and the V1 area of visual cortex in *NIgn3<sup>+/-</sup>Cyfip1<sup>+/-</sup>* females and their littermates. Scale bars, 10  $\mu$ m. \**p* < 0.05, \*\**p* < 0.01, \*\*\**p* < 0.001. The sex comparison is available in Extended Data Figure 4-1.

We investigated the impact of deleting Nlgn3 and Cyfip1 haploinsufficiency on dendritic spine density. For this purpose, we obtained  $Nlgn3^{y/-}Cyfip1^{+/-}$  double-mutant mice as well as their  $Nlgn3^{y/-}$ ,  $Cyfip1^{+/-}$ , and WT littermates, where the EGFP transgene was expressed in a subset of neurons, allowing the visualization of dendritic spines. Changes in dendritic spine density, as well as turnover, have previously been reported in the motor cortex of  $Cyfip1^{+/-}$  male mice (Bachmann et al., 2019); thus, this brain region was included in the analysis. No changes in the visual cortex were reported in these mice, hence we included this region as a control. In the motor cortex,  $NIgn3^{y/-}Cyfip1^{+/-}$  males had a significantly greater density of spines per  $10 \,\mu m$  of dendrite (mean = 2.27, SE=0.12) than  $Nlgn3^{y/-}$  males (mean=1.69, SE=0.09; Kruskal–Wallis test:  $\chi^2_{(3153)}$ =15.64, p < 0.01,  $z_{(1153)}$ = -3.27, p < 0.01) and WT males (mean = 1.66, SD = 0.13;  $z_{(1153)} = -3.41$ , p < 0.01; Fig. 4A). No significant differences in spine density were observed in the visual cortex, emphasizing the regional specificity of the effect. In contrast to the males, no significant differences in spine density in the motor or visual cortex were present in the females (Fig. 4B). To confirm the sex difference, we compared the numbers of dendritic spines in the male and female mice directly. The WT and  $Cyfip1^{+/-}$  females had on average more dendritic spines in the cortex than the WT and Cyfip1<sup>+/-</sup> males (Scheirer-Ray-Hare test, main effect of sex:  $H_{(1326)} = 98.97$ , p < 0.001; Extended Data Fig. 4-1). The selectively reduced numbers of dendritic spines in male mice raised the possibility that the social environment might impact on this phenotype, whereby being raised with their mutant littermates, WT males showed lower than expected density of dendritic spines.

# Lack of *NIgn3* and *Cyfip1* haploinsufficiency shapes the transcriptome profiles in male mice

We investigated the effect of lack of NIgn3 and Cyfip1 haploinsufficiency as well as the combined effect of both on RNA expression. Previous results raised the possibility that the social environment might affect the behavior and the spine density of the WT mice, potentially impacting on the interpretation of these findings (Kalbassi et al., 2017). To investigate the possibility that the social environment might impact on the transcriptome, we included a control group of WT males that were housed only with their WT littermates [single genotype housing (SGH)] in addition to WT males that were housed with mutant animals [mixed genotype housing (MGH)]. We performed RNA sequencing of the brain tissue, specifically the hippocampus, in *NIgn3<sup>y/–</sup>Cyfip1<sup>+/–</sup>, NIgn3<sup>y/–</sup>,* and *Cyfip1<sup>+/–</sup>* males as well as in SGH and MGH WT mice. We have selected the hippocampus as a region of interest, as it is well recognized for being sensitive to environmental changes. Moreover,

hippocampal functions have been associated with the emergence of individuality in genetically identical wildtype mice (Freund et al., 2013; Kempermann, 2019). Thus, we hypothesized that differences in the social environment of wild-type mice could affect the RNA profile in the hippocampus.

Initially, we compared the number of differentially expressed genes in the hippocampi of males from different conditions. First, we compared the mutant mice to their WT littermates (MGH WT). There were very few differentially expressed genes between the mutant mice and the MGH WT controls (Cyfip1<sup>+/-</sup> vs MGH WT: two upregulated, zero downregulated; Nlgn3<sup>y/-</sup>Cyfip1<sup>+/-</sup> vs MGH WT: three upregulated, zero downregulated; Nlgn3<sup>y/-</sup> vs MGH WT: four upregulated, 1 downregulated; Fig. 5A). The differences between the MGH and SGH WT males were, however, more substantial, with 15 upregulated and 3 downregulated genes. This difference suggests that housing conditions have the capacity to shape transcription profiles. Thus, next, we compared the mutant animals to the SGH WT controls. The differences between the SGH WT animals and the  $Cyfip1^{+/-}$  animals were still small (SGH WT vs Cyfip1<sup>+/-</sup>: 2 upregulated, 1 downregulated), while the differences between the WT males and  $Nlgn3^{y/-}$  animals were more substantial (SGH WT vs Nlgn $3^{y/}$ : 12 upregulated, 5 downregulated). There were also some differences between SGH WT controls and the NIgn3<sup>y/-</sup>Cyfip1<sup>+/-</sup> double mutants (SGH WT and NIgn3<sup>y/-</sup>  $Cyfip1^{+/-}$ : 21 upregulated, 2 downregulated), suggesting that while Cyfip1 haploinsufficiency has little impact on the transcription profile, the lack of Nlgn3 has a role in shaping it. Despite the presence of differentially expressed genes, a clear separation based on housing or the genotype of the mice was not evident in the hierarchical clustering (Fig. 5B). However, a degree of separation was present following principal component analysis, with SGH WT and MGH WT samples in particular occupying nonoverlapping space (Fig. 5C).

In the course of the analysis of a differentially expressed gene, a wealth of information about the coregulation between different genes is lost because of using an arbitrary p value level. To investigate the similarities between the transcription profiles in more depth, we used a weighted gene correlation network analysis (WGCNA). We constructed a coexpression network that contained 20 modules. Of these, the lack of *Nlgn3* was significantly associated with module 10  $[R^2 =$ -0.42, false discovery rate (FDR) = 0.09], while the reduced level of Cyfip1 was not linked to any of the modules. However, when the two mutations were combined in the  $Nlan3^{y/-}Cvfip1^{+/-}$  mutant mice, module 11 ( $R^2 = 0.66$ , FDR = 0.003), module 13 ( $R^2$  = -0.42, FDR = 0.08), and module 15 ( $R^2 = -0.43$ , FDR = 0.08) were all significantly associated with this trait, suggesting an additive effect of the two mutations (Fig. 5D). When the housing condition was considered, module 8 was found to be upregulated in the SGH animals compared with the MGH animals ( $R^2 = 0.75$ , FDR < 0.01), while module 17 was downregulated ( $R^2$  = -0.58, FDR = 0.01; Fig. 5*E*). The differential expression of the modules of genes depending on housing and genotype of the males suggests that both the social environment and the presence of *Nlgn3* and *Cyfip1* impact on the transcription profile. The genes in module 8 were associated with development, in particular with the development of the loop of Henle as well as synaptic transmission and cerebellar cell proliferation and signaling, while the genes in module 17 were associated with metabolic processes (Fig. 5*E*).

While there were some differences between the transcription profile arising from mice with different genetic mutations, the effect of housing was more pronounced.

# Social environment impacts on transcriptome profiles in male mice

We have found that WT males housed with their WT littermates have a distinctly different transcription profile from WT animals housed with mutant animals as well as the mutant animals themselves. However, the previous experiment used a cohort of WT animals from a different breeding line than the mutant animals, potentially artificially increasing the differences between the SGH and MGH WT mice. Thus, we attempted to replicate the effect of the social environment on the transcriptome in SGH and MGH WT males that came from the same breeding line and thus had the same mothers as each other as well as mutant males. Additionally, we included the single and mixed genotype housing condition for the  $NIgn3^{y/-}$  males (SGH NIgn3<sup>y/-</sup> and MGH NIgn3<sup>y/-</sup>) to investigate the potential impact of the social environment on the mutant mice. As previously, there was a fair number of genes that were differentially expressed between the MGH and SGH WT males (23 upregulated, 34 downregulated), suggesting that housing conditions shaped the transcription profiles in this group (Fig. 6A). The effect of housing was also evident to a smaller extent in the NIgn3<sup>y/-</sup> males, where there were 2 upregulated and 18 downregulated genes between SGH NIgn3<sup>y/-</sup> and MGH NIgn3<sup>y/-</sup> males. While between the SGH WT and SGH *NIgn3<sup>y/-</sup>* males that were never housed with mice of the same genotype, there were many differentially expressed genes (39 upregulated and 18 downregulated), and there were very few differences between the MGH WT and MGH NIgn3<sup>//-</sup> mice that were housed together (8 upregulates, 0 downregulated). These findings suggest that the social environment impacts the transcriptome profile not only in the WT littermates but also in the mutant  $Nlgn3^{y/-}$  mice. Despite the presence of a number of differentially expressed genes, a clear separation based on housing was not evident in the hierarchical clustering (Fig. 6B). A degree of separation was present following principal component analysis with SGH WT and MGH WT samples in particular occupying nonoverlapping space (Fig. 6C). WGCNA resulted in a network that contained 20 modules. In SGH WT animals, modules 9 and 10 were significantly upregulated ( $R^2 = 0.5$ , FDR = 0.09), and modules 14 ( $R^2 = -0.53$ , FDR = 0.08), 16  $(R^2 = -0.61, FDR = 0.04)$ , and 18  $(R^2 = 0.68, FDR = 0.02)$ were significantly downregulated (Fig. 6D). In SGH  $Nlgn3^{y/-}$  males, module 14 ( $R^2 = 0.60$ , FDR = 0.04) was significantly upregulated and module 2 was significantly downregulated ( $R^2 = -0.75$ , FDR = 0.005; Fig. 6D). Meanwhile in the MGH WT animals, modules 18 and 20 were significantly upregulated ( $R^2 = 0.55$ , FDR = 0.06; and



**Figure 5.** Nlgn3 and Cyfip1 mutations affect the transcriptome profile of male mice. *A*, Differentially expressed genes between the SGH WT, MGH WT,  $Cyfip1^{+/-}$ ,  $Nlgn3^{y/-}$ , and  $Nlgn3^{y/-}Cyfip1^{+/-}$  male mice. *B*, Hierarchical clustering of differentially expressed genes in the different housing and genotype conditions. *C*, Position of the individual samples from different housing and genotype conditions in the principal component space based on the 100 genes with the greatest fold change. *D*, WGCNA modules correlation with the different housing and genotype combinations. *E*, GO pathways for modules correlated with SGH WT condition.



**Figure 6.** Social environment affects the transcriptome profile of WT and  $Nlgn3^{y/-}$  mice. **A**, Differentially expressed genes in the SGH WT, MGH WT, SGH  $Nlgn3^{y/-}$ , and MGH  $Nlgn3^{y/-}$ . **B**, Hierarchical clustering of differentially expressed genes in the different housing and genotype conditions. **C**, Position of the individual samples from different housing and genotype conditions in the principal component space based on the 100 genes with the greatest fold change. **D**, WGCNA module correlation with the different housing and genotype combinations. **E**, GO pathways for modules correlated with SGH WT condition.

 $R^2 = 0.71$ , FDR = 0.009, respectively), and in the MGH  $Nlgn3^{y/-}$  animals, module 1 was upregulated ( $R^2 = 0.51$ , FDR = 0.09). The differential expression of the modules of genes depending on housing suggests that the social environment impacted on the transcription profile. The two modules most strongly associated with SGH WT males were module 18-containing genes responsible for cell cycle processes and chromatin regulation and module 16-containing genes associated with RNA regulation. As evident from the different GO terms associated with the SGH WT males in this cohort and the previous cohort, some of the genes in the relevant modules were different depending on the cohort.

Here we show that lack of *Nlgn3* has an impact on the RNA expression in the hippocampus, while *Cyfip1* haploinsufficiency does not. Additionally, we extend the previous finding that the presence of *Nlgn3*<sup>y/-</sup> littermates influences the social behavior of their WT littermates (Kalbassi et al., 2017) to potentially include an effect on the transcription profile. Together, these findings suggest a complex genetic–environment interaction that shapes the RNA expression.

### Discussion

In the first attempt of this kind, we investigated the combined effect of genetic and environmental factors on the phenotypes associated with ASDs. We uncovered that the *Nlgn3/Cyfip1* pathway plays a role in mouse behavior, dendritic spine density, and RNA expression. We found that sex can be a protective factor, where the females carrying the risk allele do not show the same deficits as the males. The behavior and transcriptome of WT mice is further influenced by the social group they originate from, suggesting that the social environment is an important factor modulating the phenotypes in mouse models.

As in previous literature, we found that male mice lacking Nlgn3 engage in more exploratory behavior in the open field (Radyushkin et al., 2009; Kalbassi et al., 2017) and that male mice with a reduced level of Cyfip1 expression are unable to learn motor routines (Bachmann et al., 2019). We have extended these findings to show that, unexpectedly, the accumulation of mutations can lead to a correction of the motor learning deficit. The interaction between Neuroligin3 and CYFIP1 at the synapse has been previously reported (Bachmann et al., 2019). The current findings support the idea that the consequences of this protein interaction extend beyond the molecular events, resulting in a change in the behavior of the animal. As the genetic architecture of ASDs is complex and far from being entirely understood, it is increasingly important to consider the interaction between the different proteins involved and how they shape the phenotypes seen in ASDs.

The behavioral results suggest that the nature of the functional relationship between Neuroligin3 and CYFIP1 is inhibitory. In the males with *Cyfip1* haploinsufficiency, the level of CYFIP1 was found to be reduced in several brain areas (Bachmann et al., 2019). We found that this decrease resulted in a deficit in motor learning that was accompanied by unaltered levels of activity and dendritic

spine density in the cortex (Fig. 7). This suggests that in the males heterozygous for Cyfip1, the Neuroligin3 might be binding some of the available CYFIP1 in the cellular population important for motor learning, preventing it from performing its physiological function and resulting in a deficit. In the males with NIgn3 deletion, on the other hand, physiological levels of CYFIP1 result in the mice being able to learn the motor routines. However, the lack of Neuroligin3 might result in increased levels of available CYFIP1, potentially leading to the increased activity levels seen in this model. The alternative explanation is that the increase in activity results from another pathway regulated by Neuroligin3. In the double-mutant males, which lack *NIgn3* and are heterozygous for *Cyfip1*, the increase in activity and dendritic spine density is accompanied by normalized motor learning. In this model, the level of CYFIP1 is likely reduced, but there is no Neuroligin3 to inhibit the remaining CYFIP1. As a result, there is enough CYFIP1 available to restore motor learning. Meanwhile, the lack of Neuroligin3 also results in an increased level of activity. Interestingly, only the two mutations together result in an increase of dendritic spine density. These results suggest that in the WT animals, Neuroligin3 inhibits the portion of available CYFIP1, regulating motor learning as well as potentially activity levels and dendritic spine density in the cortex.

We showed that while male mice with reduced levels of *Cyfip1* have a clear phenotype, females with the same genotype are capable of learning motor routines as usual. Additionally, our data suggest that there might be a reduction in the number of dendritic spines in the males compared with the females in the cortex. While this finding needs to be confirmed in a cohort of wild-type animals, accounting for the social dominance status, it implies that only males are affected by carrying a risk allele. While a sex difference in ASDs is frequently reported, with the affected male to female ratio being 4:1, remarkably little is known about the source of this sex difference (Halladay et al., 2015). The female protective effect is one theory aiming to explain this discrepancy, whereby females require more substantial disruption of the genetic network and the associated biological pathways to show ASD symptoms (Ferri et al., 2018). Some pathways relevant for ASDs might be redundant in females, as a number of genes associated with ASDs are X linked. While Cyfip1 is present on chromosome 7, it interacts with two X-linked genes: *NIgn3* and *Fmr1*. *Cyfip1* being part of the X-linked pathway results in unimpaired females likely because they carry two rather than one allele of the interacting genes.

In opposition to the previous report, *Cyfip1* haploinsufficiency did not affect the spine density in the motor cortex of the male mice (Bagni and Greenough, 2005; Pathania et al., 2014; Bachmann et al., 2019). While in the previous report *Cyfip1* haploinsufficiency resulted in a decrease in the dendritic density in the motor cortex, here the double-mutant mice showed an increase in dendritic spine density. The reliability of this finding is undermined by the low numbers of dendritic spines in the male mice of 1–5 spines per 10  $\mu$ m, while the numerous previous studies reported a density of 5–15 spines per 10  $\mu$ m both in the



**Figure 7.** Neuroligin3 inhibits CYFIP1 in the population of neurons important for motor learning. In  $Cyfip1^{+/-}$  males, there is a deficit in motor learning, potentially because Neuroligin3 inhibits the already reduced pool of available CYFIP1. In the  $Nlgn3^{y/-}$  males, the lack of inhibition of CYFIP1 results in motor learning, accompanied by hyperactivity. In the double-mutant males, the remaining CYFIP1 is not inhibited by Neuroligin3, resulting in the restoration of motor learning that is still, however, accompanied by hyperactivity and an increase in dendritic spine density. Thus, in the WT Neuroligin3 is likely to inhibit CYFIP1, regulating motor learning activity levels as well as dendritic spine density in the cortex.

males heterozygous for Cyfip1 and in WT males (De Rubeis et al., 2014; Pathania et al., 2014; Abekhoukh et al., 2017). This was true only for the male mice, while the sample from the female mice showed a range of 2-10 spines per 10 µm, which is in line with the previous reports. The low number of dendritic spines is unlikely to be because of the technical difficulties, as the male and female mice used in the experiment were littermates, came from the same line, and were analyzed in parallel. Therefore, it is unlikely that the low numbers in the males but not in the females are a technical artifact. The alternative explanation is that there is another factor selectively impacting the spine density in the WT males. One such factor could be the social environment. The effect of the social environment in the mouse models of ASDs has not been extensively studied. However, there were some reports of mutant animals, including males lacking Nlgn3, impacting on the behavior of the WT littermates (Yang et al., 2011; Kalbassi et al., 2017). The biological processes underlying mouse behavior are complex; however, there is a possibility that the density of dendritic spines might be correlated with behavior. Therefore, the low number of dendritic spines seen in the WT mice housed with the mutant littermates might be the result of the social environment. To confirm that the social housing effect can extend to the dendritic spine density, it would be necessary to extend the analysis to a cohort of WT males that have never been housed with mutant littermates.

In addition to the possible effect of the social environment on behavior and dendritic spine density, social housing also impacts on the transcriptome in the hippocampus. While the effect of the social environment on the gene expression was replicated in two separate cohorts of animals, the identity of the genes associated with different housing conditions varied. In the first cohort, singlegenotype housing of the WT animals was associated with synaptic transmission and metabolic processes, and in the second cohort the same trait was associated with cell cycle processes and RNA regulation. The differences might arise from the fact that one of the two cohorts of SGH WT arose from the same breeding line as the corresponding MGH WT animals while the other cohort did not. We found that wild-type individuals could potentially be differentiated based RNA expression, indicating that the different social environments have a distinct impact on wild-type mice. Although this experiment does not draw a direct link between the identity of the RNA profile and the phenotypes, we could postulate that wild-type behavior, shaped by its social environment, can be underlined by the activation of specific molecular pathways that remain to be determined.

Here we showed that not only genetic factors but also sex and social environment play a fundamental role in shaping the phenotypes of different mouse models of ASDs. Of these factors, the most complex to consider is arguably the role of the social environment. Together with a previous published study (Kalbassi et al., 2017), our

experiments suggest that the presence of mutant animals influence wild-type animals in that they adopt behavioral, cellular, and molecular phenotypic traits of mutant animals. The opposite effect, from the mutant to the wild type, seems to occur as well. The resulting effect might be that when the animals with different levels of sociability are placed in a common social context, a homogenization of their behavior, underlined by neuronal morphology and transcriptome profile changes, occurs to set an optimal level of functioning within the social group. This remains a hypothesis that needs to be validated and extended to social groups in general. But if our postulate were to be confirmed, these findings could have important implications regarding experimentation using laboratory animals and, in particular, could lead to re-evaluation of the use of wildtype animals as controls to define phenotypic traits of mouse models.

### References

- Abekhoukh S, Sahin HB, Grossi M, Zongaro S, Maurin T, Madrigal I, Kazue-Sugioka D, Raas-Rothschild A, Doulazmi M, Carrera P, Stachon A, Scherer S, Drula Do Nascimento MR, Trembleau A, Arroyo I, Szatmari P, Smith IM, Milà M, Smith AC, Giangrande A, et al. (2017) New insights into the regulatory function of CYFIP1 in the context of WAVE- and FMRP-containing complexes. Dis Model Mech 10:463–474.
- American Psychiatric Association (2012) Diagnostic and statistical manual of mental disorders. Washington, DC: American Psychiatric Association.
- Bachmann SO, Sledziowska M, Cross E, Kalbassi S, Waldron S, Chen F, Ranson A, Baudouin SJ (2019) Behavioral training rescues motor deficits in Cyfip1 haploinsufficiency mouse model of autism spectrum disorders. Transl Psychiatry 9:29.
- Bagni C, Greenough WT (2005) From mRNP trafficking to spine dysmorphogenesis: the roots of fragile X syndrome. Nat Rev Neurosci 6:376–387.
- Bariselli S, Hörnberg H, Prévost-solié C, Musardo S, Hatstatt-burklé L, Scheiffele P, Bellone C (2018) Role of VTA dopamine neurons and neuroligin 3 in sociability traits related to nonfamiliar conspecific interaction. Nat Commun 9:3173.
- Baudouin SJ (2014) Heterogeneity and convergence: the synaptic pathophysiology of autism. Eur J Neurosci 39:1107–1113.
- Bolger AM, Lohse M, Usadel B (2014) Trimmomatic: a flexible trimmer for Illumina sequence data. Bioinformatics 30:2114–2120.
- Bozdagi O, Sakurai T, Dorr N, Pilorge M, Takahashi N, Buxbaum JD (2012) Haploinsufficiency of Cyfip1 produces fragile X-like phenotypes in mice. PLoS One 7:e42422.
- Chen B, Brinkmann K, Chen Z, Pak CW, Liao Y, Shi S, Henry L, Grishin NV, Bogdan S, Rosen MK (2014) The WAVE regulatory complex links diverse receptors to the actin cytoskeleton. Cell 156:195–207.
- Chen Z, Borek D, Padrick SB, Gomez TS, Metlagel Z, Ismail AM, Umetani J, Billadeau DD, Otwinowski Z, Rosen MK (2010) Structure and control of the actin regulatory WAVE complex. Nature 468:533–538.
- Chih B, Afridi SK, Clark L, Scheiffele P (2004) Disorder-associated mutations lead to functional inactivation of neuroligins. Hum Mol Genet 13:1471–1477.
- Davenport EC, Szulc BR, Drew J, Taylor J, Morgan T, Higgs NF, López-Doménech G, Kittler JT (2019) Regulates the balance of synaptic excitation and autism and schizophrenia-associated CYFIP1 regulates the balance of synaptic excitation and inhibition. Cell Rep 26:2037–2051.
- De Rubeis S, Pasciuto E, Li KW, Fernández E, Di Marino D, Buzzi A, Ostroff LE, Klann E, Zwartkruis FJT, Komiyama NH, Grant SGN, Poujol C, Choquet D, Achsel T, Posthuma D, Smit AB, Bagni C

(2013) CYFIP1 coordinates mRNA translation and cytoskeleton remodeling to ensure proper dendritic spine formation. Neuron 79:1169–1182.

- De Rubeis S, He X, Goldberg AP, Poultney CS, Samocha K, Ercument Cicek A, Kou Y, Liu L, Fromer M, Walker S, Singh T, Klei L, Kosmicki J, Fu S-C, Aleksic B, Biscaldi M, Bolton PF, Brownfeld JM, Cai J, Campbell NG, et al. (2014) Synaptic, transcriptional and chromatin genes disrupted in autism. Nature 515:209–215.
- Dobin A, Davis CA, Schlesinger F, Drenkow J, Zaleski C, Jha S, Batut P, Chaisson M, Gingeras TR (2013) STAR: ultrafast universal RNA-seq aligner, Bioinformatics 29:15–21.
- Feng G, Mellor RH, Bernstein M, Keller-Peck C, Nguyen QT, Wallace M, Nerbonne JM, Lichtman JW, Sanes JR (2000) Imaging neuronal subsets in transgenic mice expressing multiple spectral variants of GFP. Neuron 28:41–51.
- Ferri SL, Abel T, Brodkin ES, Biomedical P, Building D, City I (2018) Sex differences in autism spectrum disorder: a review. Curr Psychiatry Rep 20:9.
- Fischer J, Hammerschmidt K (2011) Ultrasonic vocalizations in mouse models for speech and socio-cognitive disorders: insights into the evolution of vocal communication. Genes Brain Behav 10:17–27.
- Freund J, Brandmaier AM, Lewejohann L, Kirste I, Kritzler M, Krüger A, Sachser N, Lindenberger U, Kempermann G (2013) Emergence of individuality in genetically identical mice. Science 340:756–759.
- Glessner JT, Wang K, Cai G, Korvatska O, Kim CE, Wood S, Zhang H, Estes A, Brune CW, Bradfield JP, Imielinski M, Frackelton EC, Reichert J, Crawford EL, Munson J, Sleiman PMA, Chiavacci R, Annaiah K, Thomas K, Hou C, et al. (2009) Autism genome-wide copy number variation reveals ubiquitin and neuronal genes. Nature 459:569–573.
- Gumus E (2019) A hemizygous 370 kilobase microduplication at Xq13. 1 in a three-year-old boy with autism and speech delay. Fetal Pediatr Pathol 38:239–236.
- Halladay AK, Bishop S, Constantino JN, Daniels AM, Koenig K, Palmer K, Messinger D, Pelphrey K, Sanders SJ, Singer AT, Taylor JL, Szatmari P (2015) Sex and gender differences in autism spectrum disorder: summarizing evidence gaps and identifying emerging areas of priority. Mol Autism 6:5.
- Jamain S, Quach H, Betancur C, Råstam M, Colineaux C, Gillberg IC, Soderstrom H, Giros B, Leboyer M, Gillberg C, Bourgeron T (2003) Mutations of the X-linked genes encoding neuroligins NLGN3 and NLGN4 are associated with autism. Nat Genet 34:27–29.
- Jones BJ, Roberts DJ (1968) The quantiative measurement of motor inco-ordination in naive mice using an acelerating rotarod. J Pharm Pharmacol 20:302–304.
- Kalbassi S, Bachmann SO, Cross E, Roberton VH, Baudouin SJ (2017) Male and female mice lacking neuroligin-3 modify the behavior of their wild-type littermates. eNeuro 4:ENEURO.0145-17.2017–14.
- Kempermann G (2019) Environmental enrichment, new neurons and the neurobiology of individuality. Nat Rev Neurosci 20:235–245.
- Langfelder P, Horvath S (2008) WGCNA: an R package for weighted correlation network analysis. BMC Bioinformatics 9:559.
- Liao Y, Smyth GK, Shi W (2014) featureCounts: an efficient general purpose program for assigning sequence reads to genomic features. Bioinformatics 30:923–930.
- Love MI, Huber W, Anders S (2014) Moderated estimation of fold change and dispersion for RNA-seq data with DESeq2. Genome Biol 15:550.
- Martínez-Cerdeño V (2017) Dendrite and spine modifications in autism and related neurodevelopmental disorders in patients and animal models. Dev Neurobiol 77:393–404.
- Napoli I, Mercaldo V, Boyl PP, Eleuteri B, Zalfa F, De Rubeis S, Di Marino D, Mohr E, Massimi M, Falconi M, Witke W, Costa-Mattioli M, Sonenberg N, Achsel T, Bagni C (2008) The fragile X syndrome protein represses activity-dependent translation through CYFIP1, a new 4E-BP. Cell 134:1042–1054.

- Noguchi K, Gel YR, Brunner E, Konietschke F (2012) nparLD: an R software package for the nonparametric analysis of longitudinal data in factorial experiments. J Stat Soft 50:1–23.
- Pathania M, Davenport EC, Muir J, Sheehan DF, López-Doménech G, Kittler JT (2014) The autism and schizophrenia associated gene CYFIP1 is critical for the maintenance of dendritic complexity and the stabilization of mature spines. Transl Psychiatry 4:e374.
- Paxinos G, Franklin KBJ (2004). The mouse brain in stereotaxic coordinates. London: Academic.
- Phillips M, Pozzo-miller L (2015) Dendritic spine dysgenesis in autism related disorders. Neurosci Lett 601:30–40.
- Picinelli C, Lintas C, Piras IS, Gabriele S, Sacco R, Brogna C, Persico AM (2016) Recurrent 15q11. 2 BP1-BP2 microdeletions and microduplications in the etiology of neurodevelopmental disorders. Am J Med Genet B Neuropsychiatr Genet 171:1088–1098.
- R Core Team (2019) R: A language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing.
- Radyushkin K, Hammerschmidt K, Boretius S, Varoqueaux F, El-Kordi A, Ronnenberg A, Winter D, Frahm J, Fischer J, Brose N, Ehrenreich H (2009) Neuroligin-3-deficient mice: model of a monogenic heritable form of autism with an olfactory deficit. Genes Brain Behav 8:416–425.
- Rothwell PE, Fuccillo MV, Maxeiner S, Hayton SJ, Gokce O, Lim BK, Fowler SC, Malenka RC, Südhof TC (2014) Autism-associated neuroligin-3 mutations commonly impair striatal circuits to boost repetitive behaviors. Cell 158:198–212.
- Sanders SJ, He X, Willsey AJ, Ercan-Sencicek AG, Samocha KE, Cicek AE, Murtha MT, Bal VH, Bishop SL, Dong S, Goldberg AP, Jinlu C, Keaney JF, Klei L, Mandell JD, Moreno-De-Luca D, Poultney CS, Robinson EB, Smith L, Solli-Nowlan T, et al. (2015) Insights into autism spectrum disorder genomic architecture and biology from 71 risk loci. Neuron 87:1215–1233.
- Talebizadeh Z, Lam DY, Theodoro MF, Bittel DC, Lushington GH, Butler MG (2006) Novel splice isoforms for NLGN3 and NLGN4 with possible implications in autism. J Med Genet 43:e21.
- Tanaka KF, Ahmari SE, Leonardo ED, Richardson-Jones JW, Budreck EC, Scheiffele P, Sugio S, Inamura N, Ikenaka K, Hen R

(2010) Flexible accelerated STOP tetracycline operator-knockin (FAST): a versatile and efficient new gene modulating system. Biol Psychiatry 67:770–773.

- Toma C, Torrico B, Hervás A, Valdés-Mas R, Tristán-Noguero A, Padillo V, Maristany M, Salgado M, Arenas C, Puente XS, Bayés M, Cormand B (2014) Exome sequencing in multiplex autism families suggests a major role for heterozygous truncating mutations. Mol Psychiatry 19:784–790.
- van der Zwaag B, Staal WG, Hochstenbach R, Poot M, Spierenburg HA, de Jonge MV, Verbeek NE, van 't Slot R, van Es MA, Staal FJ, Freitag CM, Buizer-Voskamp JE, Nelen MR, van den Berg LH, Ploos van Amstel HK, van Engeland H, Burbach JPH (2009) A cosegregating microduplication of chromosome 15q11.2 pinpoints two risk genes for autism spectrum disorder. Am J Med Genet B Neuropsychiatr Genet 153B:960–966.
- Varoqueaux F, Aramuni G, Rawson RL, Mohrmann R, Missler M, Gottmann K, Zhang W, Südhof TC, Brose N (2006) Neuroligins determine synapse maturation and function. Neuron 51:741–754.
- Wang K, Zhang H, Ma D, Bucan M, Glessner JT, Abrahams BS, Salyakina D, Imielinski M, Bradfield JP, Sleiman PMA, Kim CE, Hou C, Frackelton E, Chiavacci R, Takahashi N, Sakurai T, Rappaport E, Lajonchere CM, Munson J, Estes A, et al. (2009) Common genetic variants on 5p14.1 associate with autism spectrum disorders. Nature 459:528–533.
- Werling DM, Geschwind DH (2013) Understanding sex bias in autism spectrum disorder. Proc Natl Acad Sci U S A 110:4868–4869.
- Yang M, Perry K, Weber MD, Katz AM, Crawley JN (2011) Social peers rescue autism-relevant sociability deficits in adolescent mice. Autism Res 4:17–27.
- Yang M, Lewis F, Foley G, Crawley JN (2015) In tribute to Bob Blanchard: divergent behavioral phenotypes of 16p11.2 deletion mice reared in same-genotype versus mixed-genotype cages. Physiol Behav 146:16–27.
- Yasuda Y, Hashimoto R, Yamamori H, Ohi K, Fukumoto M, Umeda-Yano S, Mohri I, Ito A, Taniike M, Takeda M (2011) Gene expression analysis in lymphoblasts derived from patients with autism spectrum disorder. Mol Autism 2:9.